Author's response to reviews

Title: Neurofibromatosis type 1 (von Recklinghausen's Disease) with duodenal GIST that was difficult to distinguish from pancreatic neuroendocrine tumor: a case report

Authors:

Shinji Ohtake (56715203@ezweb.ne.jp)
Noritoshi Kobayashi (norikoba@yokohama-cu.ac.jp)
Shingo Kato (shingo800m@hotmail.com)
Takashi Uchiyama (maruru0203@yahoo.co.jp)
Hironori Mawatari (mawatari-ykh@umin.ac.jp)
Hiroshi Iida (hiro19780905@yahoo.co.jp)
Hiroki Endo (t066011b@yokohama-cu.ac.jp)
Yuichi Nozaki (nzaki-y@fukuhp.yokohama-cu.ac.jp)
Kunihiro Hosono (hiro1017@fukuhp.yokohama-cu.ac.jp)
Koji Fujita (kfujita@yokohama-cu.ac.jp)
Masato Yoneda (yoneda@med.yokohama-cu.ac.jp)
Hirokazu Takahashi (hirokazu@med.yokohama-cu.ac.jp)
Masahiko Inamori (inamorim@med.yokohama-cu.ac.jp)
Yasunobu Abe (a0121@yokohama-cu.ac.jp)
Hiroyuki Kirikoshi (hkirkosh@med.yokohama-cu.ac.jp)
Kensuke Kubota (kubotak@yokohama-cu.ac.jp)
Satoru Saito (ssai1423@yokohama-cu.ac.jp)
Itaru Endo (endoito@med.yokohama-cu.ac.jp)
Yoshiaki Inayama (y-ina@fukuhp.yokohama-cu.ac.jp)
Atsushi Nakajima (nakajima-tky@umin.ac.jp)

Version: 2 Date: 30 January 2010

Author's response to reviews: see over
Editorial Office

Jan. 30 2010

Dear Editors

We are pleased to submit our manuscript entitled “A case of neurofibromatosis type I (von Recklinghausen’s Disease) with a duodenal gastrointestinal stromal tumor that was difficult to distinguish from a pancreatic neuroendocrine tumor” for publication consideration in “Journal of Medical Case Report.” Among the NF-1 associated GISTs, duodenal GISTs are particularly difficult to distinguish from periampullary endocrine tumors, not only in terms of the radiological findings, but also in terms of the etiological features. We believe that our paper would be of great interest to the readers of “Journal of Medical Case Report” and look forward to hearing your response.

Sincerely yours,

Noritoshi Kobayashi, M.D., Ph.D.
Gastroenterology Division, Yokohama City University Hospital
3-9, Fuku-ura, Kanazawa-ku, Yokohama, 236-0004, Japan.
TEL: +81-45-787-2640, FAX: +80-45-784-3546
E-mail: norikoba@.yokohama-cu.ac.jp